



Industrial policy and global public goods provision: rethinking the environmental trade agreement

Pia Andres

January 2023

Centre for Climate Change Economics and Policy Working Paper No. 413 ISSN 2515-5709 (Online)

Grantham Research Institute on **Climate Change and the Environment** Working Paper No. 388 ISSN 2515-5717 (Online)



Economic and Social Research Council

THE GRANTHAM FOUNDATION







The Centre for Climate Change Economics and Policy (CCCEP) was established by the University of Leeds and the London School of Economics and Political Science in 2008 to advance public and private action on climate change through innovative, rigorous research. The Centre is funded by the UK Economic and Social Research Council. Its third phase started in October 2018 with seven projects:

- 1. Low-carbon, climate-resilient cities
- 2. Sustainable infrastructure finance
- 3. Low-carbon industrial strategies in challenging contexts
- 4. Integrating climate and development policies for 'climate compatible development'
- 5. Competitiveness in the low-carbon economy
- 6. Incentives for behaviour change
- 7. Climate information for adaptation

More information about CCCEP is available at www.cccep.ac.uk

The Grantham Research Institute on Climate Change and the Environment was established by the London School of Economics and Political Science in 2008 to bring together international expertise on economics, finance, geography, the environment, international development and political economy to create a world-leading centre for policy-relevant research and training. The Institute is funded by the Grantham Foundation for the Protection of the Environment and a number of other sources. It has 13 broad research areas:

- 1. Biodiversity
- 2. Climate change adaptation and resilience
- 3. Climate change governance, legislation and litigation
- 4. Climate, health and environment
- 5. Environmental behaviour
- 6. Environmental economic theory
- 7. Environmental policy evaluation
- 8. International climate politics
- 9. Science and impacts of climate change
- 10. Sustainable public and private finance
- 11. Sustainable natural resources
- 12. Transition to zero emissions growth
- 13. UK national and local climate policies

More information about the Grantham Research Institute is available at www.lse.ac.uk/GranthamInstitute

Suggested citation:

Andres P (2023) Industrial policy and global public goods provision: rethinking the environmental trade agreement. Centre for Climate Change Economics and Policy Working Paper 413/Grantham Research Institute on Climate Change and the Environment Working Paper 388. London: London School of Economics and Political Science

This working paper is intended to stimulate discussion within the research community and among users of research, and its content may have been submitted for publication in academic journals. It has been reviewed by at least one internal referee before publication. The views expressed in this paper represent those of the authors and do not necessarily represent those of the host institutions or funders.

Industrial Policy and Global Public Goods Provision: Rethinking the Environmental Trade Agreement*

Pia Andres[†]

January 18, 2023

Abstract

Countries around the world increase the downstream cost of low carbon technologies using anti-dumping duties and local content requirements, while simultaneously blaming inadequate efforts to address climate change on the economic cost of doing so. This paper presents a 2-country, 2-period strategic model of trade in a clean technology in the presence of differential country-level production costs and imperfect competition. If the difference in production cost is large enough and learning-by-doing allows the laggard country to catch up, then in the absence of production subsidies remaining in autarky during Stage 1 of the game can be welfare-improving for both countries. This result is strengthened when both countries use consumer subsidies. When countries choose their policy mixes, the Nash Equilibrium involves both trade and production subsidies on the part of the laggard country. The analysis suggests that an environmental trade agreement is most likely to be beneficial if production subsidies for clean technology are explicitly permitted.

^{*}I am grateful to Scott Barrett, Peter Burr, Simon Dietz, Eugenie Dugoua, Ben Filewood, Roger Fouquet, Mads Greaker, Bård Harstad, David Hémous, Katinka Holtsmark, Antony Millner, Giorgio Presidente, and Misato Sato for their thoughtful comments at various stages of this paper's development, as well as the participants of the seminars and workshops where this paper was presented and discussed. All remaining errors and omissions are mine. I acknowledge financial and administrative support from the Fulbright Commission in Belgium, the Grantham Research Institute on Climate Change and the Environment, and the Economic and Social Research Council through the Centre for Climate Change Economics and Policy.

[†]Grantham Research Institute and Department of Geography and Environment, London School of Economics and Oxford Martin School, University of Oxford. Email: p.andres@lse.ac.uk

1 INTRODUCTION

Climate change has been an item on the agenda of global diplomacy and politics for decades. Policy action at the level of nation states, however, has to date remained woefully insufficient to achieve stated aims to reduce greenhouse gas emissions and avoid catastrophic warming. The reason for this is, according to much of the economics literature, the cost of reducing emissions – and the fact that due to the global nature of climate change, each country has an incentive to free-ride on others' efforts (cf. Barrett 1994a, 2005; Finus 2008). One might therefore expect that reductions in the cost of technological solutions would be universally welcomed. And yet, policy support for renewable energy, which is a key component of a climate-compatible global production system, has frequently been accompanied by or met with trade barriers, which are associated with increased user costs.¹ Such decisions are taken in the interest of each country's domestic industry producing renewable energy technology.

Are there conditions under which this might prove beneficial to climate action? The literature on industrial policy and infant industry maturation tends to focus on the effectiveness of industrial policy and the cost-benefit trade-off at the level of the country promoting its infant industry (e.g. Krueger et al. 1982; Melitz 2005; Young 1991). Here, I analyze the global as well as the local welfare implications of allowing an infant industry to mature in the context of a product with positive consumption externalities. The analysis suggests that such externalities could make infant industry maturation beneficial for both the initially laggard and the initially frontier country. Conversely, in the absence of positive consumption externalities, the frontier country prefers to extract rents from the laggard country's consumers.

China's entry into the solar photovoltaics (hereafter PV) market, which was enabled by aggressive industrial policy, considerably increased competition in this market² and is widely credited to have contributed to the dramatic declines in the cost of solar PV energy over the past decade (cf. Carvalho et al. 2017; Dent 2018; Nemet 2019). This raises the following question: if industrial policy – possibly including temporary protectionist measures – is necessary to allow a potential supplier country of clean technology to realize its potential and increase competition in the future, can the long-term global benefits of doing so outweigh the short-term cost of protection? In this paper, I present a stylized model highlighting conditions under which this may be the case. I then add the use of consumer or producer subsidies, and show how this alters the trade-offs considered. The key insight is that allowing the initially laggard country to more effectively compete with the frontier country can be beneficial for the

^{1.} For example, China required 40% of wind turbines and blades to be manufactured locally for projects to be eligible for public tenders as early as 1997. Between 2011 and 2017, India imposed a 60% local content requirement on tenders for solar photovoltaics (PV), and 30% for concentrated solar power. France's feed-in tariff for solar PV between 2002 and 2012 came with a 60% local content requirement (Scheifele et al. 2022).

^{2.} In 2008, the 10 largest solar PV equipment manufacturers accounted for almost 90% of global market share, operating in just four countries (Germany, the US, Switzerland and Japan). By 2021, the top ten manufacturers' share had dropped by half due to new firm entry. Today, all top ten equipment manufacturers are in China and claim over 45% of the global market share (IEA 2022).

global economy, and allowing for the use of producer subsidies can accomplish this goal while avoiding the temporary cost of protectionism.

I model the production and consumption of an environmentally beneficial product in a 2country, 2-stage game with imperfect competition, differences in initial production cost, and learning-by-doing. Each country has a representative firm which can produce a technology capable of generating private as well as public benefits. We can think of this as a renewable energy technology which benefits consumers directly by supplying energy, as well as reducing negative externalities which would result if fossil fuels were used instead. While the paper is motivated by the global challenge of addressing climate change, the model is potentially relevant to any game in which consuming a technology carries benefits to other consumers at home and abroad. Other examples include the use of vaccines, which also generate both private and public benefits (cf. Brito et al. 1991; Fisman et al. 2009; Francis 1997).

I start with the assumption that there are no subsidies and governments can choose only whether to trade or not. Firms play a Bertrand game if countries engage in trade, and act as local monopolists otherwise. If the laggard (high cost) firm is active during Stage 1, it 'catches up' with the frontier (low cost) firm by Stage 2. If countries trade, then the laggard firm remains inactive and the frontier country extracts rents from the laggard country's consumers. Under both trade and autarky and in the absence of other policy action, consumption and public goods provision are inefficiently low even without the global public goods aspect due to imperfect competition.³

If the laggard firm catches up, Bertrand competition leads to marginal cost pricing in Stage 2. The model shows how for a sufficiently large difference in initial marginal cost, and in the absence of other policy tools, autarky in Stage 1 can benefit both countries through increased consumer surplus and public goods provision in Stage 2. The frontier country loses the rents it could otherwise extract from the laggard country, which may yet make infant industry maturation a net reduction in welfare for the frontier country.

In the presence of market failures, such as those arising from the positive externality and imperfectly competitive market structure considered here, a social planner will usually wish to intervene beyond the decision of whether or not to trade. Indeed, various subsidies, preferential loans and other incentives implemented by policy-makers around the world have been decisive in scaling up the deployment of renewable energy technologies.

I therefore analyze how quantity subsidies affect the dynamics of trade and public goods provision. The model suggests that when consumer subsidies are available as a policy tool, temporary autarky is beneficial for any distance to frontier, as competition from the laggard country in Stage 1 no longer provides benefits in terms of constraining the frontier monopolist (for the frontier country this is once again counterbalanced by the loss in rents). Producer subsidies exacerbate free rider problems and are generally undesirable as the main climate policy. Under trade they allow the laggard country to free ride on the frontier country's subsidies, as

^{3.} Arguably a reasonable assumption for a new industry, see e.g. Fischer et al. 2017, 2018

only the frontier firm is active, and make autarky the preferred choice from the perspective of public goods provision.

If countries are free to choose their policy mix then the laggard country will choose a producer subsidy which is sufficiently high to enforce marginal cost pricing, but which leaves the frontier firm to supply the whole market. Both countries subsidize consumers to the point of internalizing the domestic part of the public good. The outcome is equivalent to the noncooperative equilibrium obtained under perfect competition. While the first best is not attainable, the 'second best' policy mix which achieves an outcome equivalent to a game with perfect competition thus requires that countries are able to choose a mix of consumer and producer subsidies.

The paper connects the literature on industrial policy and infant industry protection (cf. Chang 2003; Krueger et al. 1982; Melitz 2005; Young 1991) to that highlighting increased competition as a cause of gains from trade (Krugman 1979; Markusen 1981) by modelling learning-by-doing in an initially laggard industry as a pre-condition to enable competition later on. It also builds on the literature on international collective action problems related to climate change and the environment (see e.g. Barrett 1994a, 2005; Finus 2008; Harstad 2016). The positive global externality arising from consumption of the good implies that the benefits of an infant industry catching up may be global, rather than being limited to the initially laggard country. In contrast, in the absence of positive externalities the advanced country always prefers to retain its ability to extract rents, despite associated losses in consumer surplus.

Previous work on the infant industry argument has emphasized the need for the cost of temporary protection to be outweighed by the benefits of domestic production in a higher valueadded industry later (cf. Krueger et al. 1982); for example, because learning-by-doing exhibits spillovers across goods (Young 1991) or domestic and foreign goods are imperfectly substitutable (Melitz 2005). This paper highlights an additional channel through which infant industry protection may be warranted. I model the maturation of an infant industry as a precondition for allowing gains from trade via increased competition (Krugman 1979; Markusen 1981) to increase in the future.

The model allows me to identify conditions for infant industry protection to improve welfare in the initially laggard country, as well as consumer surplus in both the laggard and the frontier country, even while abstracting from any macroeconomic spillovers or growth effects. I also highlight how the frontier country can, under trade, extract rents from the laggard country, offsetting any losses in consumer surplus from imperfect competition. Finally, introducing a global positive consumption externality implies that infant industry protection can be welfare improving not only for the initially laggard country, but also for the frontier country. The larger the public benefit from consumption becomes relative to the private benefit, the closer countries' incentives with respect to trade should align.

Efforts to liberalize trade in 'green' technologies – including, but not limited to, those with the potential to reduce greenhouse gas emissions – have been underway for years. In the 2001

Doha declarations ministers stated their commitment to negotiations on reducing or eliminating tariff and non-tariff barriers to environmental goods and services (Balineau et al. 2013; Droege et al. 2016). The Asia-Pacific Economic Cooperation (APEC) countries reached an environmental trade agreement in 2012 (Jacob et al. 2017; Steenblik 2005; Vossenaar 2016), while negotiations for a World Trade Organization (WTO)-wide agreement are ongoing (De Melo et al. 2020; Monkelbaan 2017). The theoretical rationale for liberalizing trade in clean technologies is clear: doing so is expected to facilitate diffusion of such technologies, thereby increasing their deployment, and enabling greater climate change mitigation and other environmentally beneficial outcomes at a lower cost. In practice, however, countries' attitudes towards these technologies have frequently proven to be mercantilist in nature (cf. De Melo et al. 2022).

This paper suggests that while a trade agreement may be beneficial for climate action and global welfare, producer subsidies should be allowed where they may level the competitive playing field, and more generally whenever countries choose to use them. It further provides intuition for why countries which provide consumer subsidies for renewables often use local content requirements, as well as why early mover countries such as the US or Germany tend to oppose producer subsidies in other countries – even at the expense of their own consumers.⁴

The remainder of the paper proceeds as follows. Section 2 summarizes the literatures on trade and international public good games which the paper builds on. Section 3 introduces the key tenets of the model and the benchmark 'first best' cooperative outcome, comparing it to a status quo under which countries are in autarky and do not subsidize the technology in any way. Section 4 analyzes the welfare implications of trade when no other climate policy is available, and identifies the conditions under which autarky may be individually or jointly preferable to trade. Section 5 introduces consumer and producer subsidies and analyzes how this changes the dynamics of the game, both separately and when countries are free to choose their subsidy mixes. Section 6 discusses the evolution of the solar PV sector as a real-world example of the dynamics the model seeks to highlight. Section 7 concludes.

2 RELATED LITERATURE

Gains From Trade and Infant Industries The trade literature identifies many mechanisms through which gains from trade may materialize. These include the efficiency gains of each country specializing where it has a comparative advantage (Ricardo 1891); increased competition and increasing returns to scale in a larger market (Krugman 1979); and a redistribution of market share towards the most productive firms and the exit of the least productive (Baldwin et al. 2004; Melitz 2003). A larger potential market might further increase incentives to innovate

^{4.} The expansion of low-cost solar panel manufacturing in China was met with anti-dumping duties by both the United States and the European Union (Hughes et al. 2017; Meckling et al. 2018; Wu et al. 2013). Estimates of the cost of US protective tariffs downstream, both in the solar PV sector and more broadly, include Houde et al. 2022 and Fajgelbaum et al. 2020.

(Aghion et al. 2018; Grossman et al. 1990) and raise the potential for knowledge spillovers and technology diffusion (Grossman et al. 1990; Keller 2004), by making technology more widely and cheaply available (Carbaugh et al. 2012; ICTSD 2011).

If comparative advantage and industrial structure are taken as fixed, the benefits of liberalizing trade are clear and highly intuitive. However, patterns of comparative advantage are not solely determined by fundamental endowments (cf. Hausmann et al. 2007). In highly complex modern industries in particular, competitiveness is developed over time. Many scholars argue that industries need temporary protection from import competition in order to develop and become competitive (Chang 2003; Hanlon 2017; Juhász 2018). This is known as the 'infant industry argument'. Infant industry protection is usually seen as a strategy for developing countries, but can also play a role in building productive capabilities in developed economies, especially if the industry in question is underdeveloped in a particular country (Andreoni et al. 2016).

Theoretical models suggest that temporary protection can be beneficial when entry barriers and dynamic learning effects are high (Irwin 2000; Melitz 2005; Young 1991). The temporary costs of protecting the infant industry must be outweighed by the benefits of domestic production in a higher value-added industry later on (Krueger et al. 1982). Empirical evidence on the justifications for and effectiveness of infant industry protection are mixed: Krueger et al. 1982 show that protected industries in Turkey over the period 1963-1976 did not experience faster cost declines than others, and argue that infant industry protection could therefore not be a valid argument for the use of tariffs. Conversely, Hanlon 2017 argues that competition from Britain hindered North American shipbuilders' ability to transition from wood to metal shipbuilding in the late 19th century, while Juhász 2018 shows that the blockade of British imports during the Napoleonic wars enabled more protected French regions to more rapidly transition to mechanized cotton spinning.

Overall, more competition through trade may not necessarily be beneficial during the early stage of developing a new industry. This may provide some justification for the use of instruments such as local content requirements (cf. Johnson 2016). The returns from infant industry protection are usually modelled as greater future growth via the reallocation of output to more rapidly growing industry, inter-industry spillovers enabling learning-by-doing (Young 1991), the result of imperfect substitutability between domestic and foreign goods (Melitz 2005), or protection against a sudden demand-shock favouring a foreign-produced good with non-linearly increasing production cost (Traiberman et al. 2022). In contrast, this paper explores the implications of infant industry dynamics for consumer surplus and public goods provision under trade and autarky, and in particular the potential global benefits of temporarily protecting an infant industry through its impact on competition later on. I thereby identify an additional mechanism through which infant industry protection may be beneficial, and explicitly model the rents which the initially frontier country can extract from the initially laggard country under imperfect competition in the absence of any policy supporting the infant industry. The

paper does not explicitly model the source of these dynamics, nor does it consider other, more long-term, implications of competition and market size beyond prices and quantities, such as innovation or learning by the frontier industry.

Climate Change and International Cooperation Climate change is an inherently global problem, which must nevertheless be addressed within the current framework of individual nation-states. Action on climate change and other transboundary environmental problems involves strategic interaction between individual countries, which makes game theory an attractive tool of analysis. In the absence of a supra-national authority which could force countries to act to achieve the global social optimum, incentives to free ride on others' efforts abound, making it extremely difficult for international cooperation to be achieved.

A broad literature has therefore used game theory to formally analyze the mechanisms at play in international climate negotiations. Due to the absence of an authority which could hold countries to a binding agreement, non-cooperative game theory is usually thought most relevant (Barrett 2005; Finus 2008). This literature typically attempts to provide insights on how treaties may improve on the status quo, using two benchmark cases: no agreement with each country only taking into account its own environmental damages and ignoring the transboundary externalities caused by its emissions; and the global 'first-best' or fully cooperative outcome, which would be obtained if a benevolent social planner could set global policy (e.g. Barrett 1994a; Battaglini et al. 2016; Harstad 2012).

Technology as a potential channel for enhancing international cooperation has also been explored. Barrett 2006 discusses if and how a system of two treaties promoting R&D and adoption of a resulting breakthrough technology could enhance cooperation. He argues that the R&D and technology approach faces the same challenges as the Kyoto approach, with the exception of breakthrough technologies with increasing returns to scale. Building on Barrett 2006, Hoel et al. 2010 show that when R&D costs affect adoption costs, a large stable coalition is possible and can improve welfare. Harstad 2016, however, points out that in a dynamic setting, green investment may be negatively affected by the hold-up problem identified in earlier literature (cf. Beccherle et al. 2011): incentives to invest in green technology may be reduced if countries expect this will force them to agree to abate more in future negotiations, which is especially damaging in the presence of technological spillovers. Conversely, Battaglini et al. 2016 present a dynamic model with incomplete contracts, in which the non-contractibility of investments in green technology can help leverage the hold-up problem when agreement duration is endogenous: in their model, a short-term agreement with low investment is used as a credible threat against free-riding, bringing about a longer-term, more comprehensive agreement.

This paper also relates to a growing body of literature exploring international environmental or climate cooperation in the presence of international trade. Research in this area typically explores the issue of pollution leakage and the potential for border adjustments (e.g. Barrett 1994b; Grubb et al. 2022; Richter et al. 2021); the potential use of trade policy to incentivize cooperation, also referred to as 'issue linkage' (e.g. Barrett 1997; Barrett et al. 2022; Hagen et al. 2021; Nordhaus 2015); or both (e.g. Helm et al. 2012). The paper departs from most existing research on trade and the environment in that it considers trade in pollution-reducing, rather than polluting, products.⁵ Existing work in this vein includes Fischer et al. 2017, who compare the relative merits of up- and downstream subsidies when regions set different emission taxes and upstream producers engage in Cournot competition, selling abatement technology to downstream subsidies, as a downstream subsidy increases the global price of abatement technology, leading the other region to use less of it.⁶

3 THE MODEL

Assume that a climate friendly technology has been developed. Using this technology instead of fossil fuels reduces climate damages and thus negative externalities. I abstract from the use of fossil energy and model the technology as generating positive global externalities. Two countries are facing climate change. Each country has a domestic industry able to produce some quantity q of the technology at constant marginal cost c if it has experience producing it in the previous period, and dc otherwise, where d > 1. This is captured by

$$c_{i,t}(q_{i,t}) = \begin{cases} cq_{i,t} & \text{if } q_{i,t-1} > 0\\ dcq_{i,t} & \text{if } q_{i,t-1} = 0 \end{cases}$$
(1)

The game has two stages, both of which are identical. For simplicity, assume there is no discount factor. To aid readability, time subscripts will be dropped wherever possible.

Downstream Consumer Recall that consuming the technology generates both private and public benefits. In each stage of the game, the representative consumer in country *i* chooses how much r_i to consume in order to maximize utility $U(r_i, Y)$ subject to prices and budget constraints, without taking into account the positive externalities generated. We can think of this as a consumer setting the marginal private benefit from consuming a cleaner technology (such as the utility gained from electricity) equal to the price, without taking into account that

^{5.} In reality, of course, the production process itself of so-called 'clean technology' is rarely carbon-neutral. However, for the purposes of this analysis I will focus on the mitigation potential of a clean technology and its resultant positive externalities.

^{6.} The implications of upstream versus downstream subsidies identified in this paper contrast with those found by Fischer et al. 2017, whose set up differs from that presented here in several respects. Fischer et al. 2017 model pollution taxes as the baseline environmental policy and include subsidies as an additional policy to address leakage, which is unilaterally set in the region in which pollution taxes are higher. This paper abstracts from the polluting sector and models the benefits of clean technology as a public good, making subsidies the only environmental policy. Bertrand competition implies that prices always equal the laggard firm's constant marginal cost and are not affected by demand. Finally, I consider the Nash equilibrium in subsidies, creating incentives for each country to free-ride on the other's upstream subsidies, which due to rent dissipation under Bertrand competition yield no additional benefits via increased market share.

using a dirtier technology (such as coal) instead would cause climate damages borne by all. Consumers in country *j* similarly choose r_j . Assuming that utility from the technology and the numeraire good Y are additive, in each stage the consumer solves

$$\max_{\{r_i\}} U_i(r_i, Y_i) = u(r_i) + Y_i$$
(2)

subject to the budget constraint⁷

$$Pr_i + Y_i = M_i \tag{3}$$

This yields the demand function

$$P = \frac{\delta u_i}{\delta r_i} \tag{4}$$

I assume that utility is continuously increasing in r_i , concave and twice differentiable – hence $\frac{\delta u}{\delta r_i} > 0$ and $\frac{\delta^2 u}{\delta r_i} < 0$. Utility and demand in country *j* are symmetric. The positive externality from consumption, which I will refer to as the global public good, is a linear function of $r_i + r_j = R$.

Firms Each country has a representative firm which maximizes profits during each stage of the game. Under autarky, firm *i* maximizes

$$\max_{\{q_i\}} \prod_i = P(q_i)q_i - cq_i \tag{5}$$

I assume that country j is a laggard who has not produced the technology in period 0, such that

$$\max_{\{q_j\}} \prod_j = P(q_j)q_j - dcq_j \tag{6}$$

Under trade, price is instead a function of $r_i + r_j = R.^8$

Both Cournot and Bertrand competition are highly stylized models of oligopolistic competition. Here, Bertrand competition simplifies the analysis without sacrificing any central insights the paper seeks to provide.⁹ Firms therefore play a Bertrand game in which the low cost firm *i*

^{7.} Note that by assumption, there is no income effect affecting the demand for the good. While this is not a realistic assumption, it is a helpful simplification which allows us to focus on the implications of countries' differential production costs for potential gains from trade for the particular market under consideration, while abstracting from other country-level differences or macroeconomic considerations.

^{8.} However, since both consumers have the same demand function this makes no difference to the slope of market demand.

^{9.} Under Cournot competition, the maturation of an infant industry would still increase competition and reduce prices, but the global market would only approach the competitive equilibrium once a sufficiently large number of firms (clearly exceeding the two firms considered here) enters. The key dynamics illustrated in this paper would likely still hold, but presenting them would require a more complex model with a more sophisticated learning curve. Moreover, given the context which motivates this paper, in which Chinese entrants into the solar PV market

supplies the whole market at $P = dc - \varepsilon$. Note that this requires that *d* is sufficiently small for dc to be below the monopoly price. Also implicit in the analysis is the assumption that

$$\frac{\delta u_i}{\delta r_i}(0) > dc \tag{7}$$

which is a requirement for consumption to take place in both countries under all scenarios considered, including in the absence of trade or subsidies.

Government The government takes into account cumulative welfare over both stages of the game. It chooses subsidies *s* and whether or not to engage in trade, taking into account consumer surplus *CS*, firm profits Π , the benefit generated by consuming the technology which is assumed to be linear, and the actions of the other government. Prices are assumed to be set based on supply, demand and market power. Trade requires both countries' agreement, such that it is sufficient for one country to prefer autarky to prevent trade.¹⁰ Both governments are assumed to have identical objective functions, maximizing cumulative national welfare which equals

$$\Sigma_{t=1}^2 W_{i,t} = \Sigma_{t=1}^2 CS_{i,t}(s_{i,t}, s_{j,t}) + \Pi_{i,t}(s_{i,t}, s_{j,t}) + \frac{b}{2} R_t(s_{i,t}, s_{j,t}) - cost(s_{i,t})$$
(8)

where in each stage t of the game, the private (national) benefit from global consumption of R_t is $B_{i,t} = \frac{b}{2}R_t$ and the global benefit from global consumption of R_t is $B_t = bR_t$. The overall cost to country *i* of subsidizing is $cost(s_i)$. $W_{i,t}$ denotes overall welfare in country *i* in stage t of the game.

3.1 First Best

The global optimum requires that global marginal benefits from consuming the technology equal the lowest attainable marginal cost. Thus,

$$\frac{\delta u}{\delta r_i} + b = c \tag{9}$$

in each stage of the game. A number of subsidy mixes could in principle achieve this outcome, each of which has different distributional implications. If both countries use consumer subsidies, then since P = dc as a result of Bertrand competition, the market outcome (with a downstream quantity subsidy) will be

sold their products on the world market at considerably lower prices than those previously available from western incumbents (for more detail, see section 6), I argue that Bertrand competition is a reasonable modelling choice.

^{10.} In practice, of course, trade policy tends to focus either on import or export barriers. Strategic sectors might be protected from import competition, while key inputs into such sectors might be subject to export restrictions. However, trade barriers do tend to be reciprocal where possible, and free trade agreements require mutual cooperation. I therefore consider this a reasonable simplification.

$$\frac{\delta u}{\delta r_i} + s = dc \tag{10}$$

and the optimal subsidy satisfies

$$s = c(d-1) + b \tag{11}$$

If the subsidy is shared equally in the form of a consumer subsidy, then the low cost country extracts rents from the high cost country amounting to $c(d-1)r_j$. This is not an equilibrium, as in both stages of the game the high cost country will wish to set marginal local benefit equal to marginal local cost. Since in country *j* the latter remains equal to P = dc, this optimization problem yields $s_j = \frac{b}{2}$. A producer subsidy borne by the low cost country implies that the low cost country subsidizes the high cost county with br_j , thus bearing the full cost of public goods provision. While the low cost country will wish to subsidize foreign consumers to the extent that this increases local public goods provision, it will not account for increases in foreign consumer surplus or foreign public goods provision (see Appendix B.2). Neither option can thus constitute a Nash equilibrium. Alternatively, a producer subsidy amounting to $c(d-1) - \varepsilon$ set by the high cost country could enforce marginal cost pricing by the low cost firm, while a consumer subsidy of *b* could be shared between both countries. Can this more equitable solution be a stable equilibrium?

In Stage 2, each country will have the same welfare which equals

$$W_{i} = u(r_{i}) + br_{i} - cr_{i} + \frac{b}{2}r_{i} + \frac{b}{2}r_{j} - br_{i}$$
(12)

Each country will have an incentive to unilaterally deviate and set the subsidy equal to $\frac{b}{2}$, since individual country welfare is optimized if the marginal cost of consumption equals its marginal (domestic) benefit, which is at $\frac{\delta u}{\delta r_i} + \frac{b}{2} = c$. Since countries foresee a deviation in Stage 2, cooperation cannot be sustained in Stage 1 either.

3.2 Business As Usual

Suppose that under business-as-usual, countries are in autarky and no climate or industrial policy is used. Each firm acts as a monopolist, such that in both Stage 1 and Stage 2, in country i,

$$\frac{\delta P}{\delta q_i} q_i + P(q_i) = c$$

$$\rightarrow \frac{\delta^2 u}{\delta r_i} r_i + \frac{\delta u}{\delta r_i} = c$$
(13)

Since $\frac{\delta^2 u}{\delta r_i} < 0$ this outcome is clearly inferior from the consumer's point of view than a

competitive equilibrium, where $\frac{\delta u}{\delta r_i} = c.^{11}$ In country *j*,

$$\frac{\delta P}{\delta q_j} q_j + P(q_j) = dc$$

$$\rightarrow \frac{\delta^2 u}{\delta r_j} r_j + \frac{\delta u}{\delta r_j} = dc$$
(14)

in Stage 1. In Stage 2, country j has caught up with country i, and both firms set prices as in equation 13.

While we saw in section 3.1 that the global optimum may not be attainable, the status quo can be improved upon using a set of policies. In the following, I will analyze equilibrium outcomes and pay-offs under different trade and subsidy regimes, evaluate what kind of 'game' countries will choose to opt into if given the option, and propose a 'second best' policy mix which allows for the outcome which is closest to the first best.

4 TRADE AND AUTARKY WITHOUT SUBSIDIES

Can a trade agreement constitute an improvement over business-as-usual? In principle, trade allows both countries to benefit from the more advanced industry's lower marginal cost. Under perfect competition, this is usually a straightforward welfare improvement (see Appendix C). However, in practice an emerging new industry is unlikely to be competitive, especially one which depends on government support for its viability. Section 6 argues that in case of solar panels, a few powerful players were highly dominant until China's entry into the global market. The analysis therefore assumes that firms are imperfectly competitive, and that entry from the initially high-cost country is required to challenge the monopoly power of the technological leader. I will use backward induction to analyze whether countries will find it beneficial to trade in each stage.

In Stage 2, the low cost firm will set $P = dc - \varepsilon$ if the high cost firm still has marginal cost dc, and supply the whole market. If the high cost firm has caught up, both firms share the market at P = c. Assuming once again that d is sufficiently low for dc to be below the monopoly price, it is trivial to show that consumer surplus, consumption, and the public good are all higher under trade in either case.

Suppose the high cost country has not caught up. Then, $\frac{\delta u_i}{\delta r_i} = \frac{\delta u_j}{\delta r_j} = dc$. Moreover, the high cost country extracts rent from the high cost country amounting to $c(d-1)r_j$.

Conversely, if the high cost country has caught up, $\frac{\delta u_i}{\delta r_i} = \frac{\delta u_j}{\delta r_j} = c$. Since c < dc, prices are lower, consumption, consumer surplus and the public good are higher, and the deadweight loss of monopoly has been eliminated. However, the low cost country is no longer able to extract rent from the high cost country, implying that it is ambiguous under which scenario its domestic

^{11.} $c - \frac{\delta^2 u}{\delta r_i} r_i > c$. Under monopoly, prices are higher and consumption, consumer surplus and public goods provision lower.

welfare is higher. In either case, however, neither country wishes to be in autarky in Stage 2.

In Stage 1, both countries face the same contemporaneous trade-off: welfare is higher under trade than autarky. However, recall that equation 1 implies that the high cost firm will have the same marginal cost as the low cost firm in Stage 2 if it is active in Stage 1. If countries trade, the low cost firm supplies the whole market during both periods. If they remain in autarky, the high cost country – and possibly the low cost country, too – trade off welfare losses in Stage 1 for welfare gains in Stage 2.

At this point, it is useful to parameterize the model. Suppose the utility function is of the form

$$u(r_i) = ar_i - \frac{r_i^2}{2}$$
(15)

yielding the following demand and inverse demand functions

$$P = a - r_i$$

$$r_i = a - P$$
(16)

The assumption that dc is below the low cost firm's monopoly price becomes

$$\frac{a+c}{2c} > d \tag{17}$$

Appendix A derives individual payoffs and welfare implications for both countries under Stage 1 autarky and Stage 1 trade for the utility function specified in equation 15. It leads us to

Proposition 1. If the high cost firm's distance to frontier is sufficiently large, and if protection from trade allows it to catch up, then temporary autarky can increase overall public goods provision for both countries.

Proof: Result 1, Appendix A.

Corollary 1. If the high cost firm's distance to frontier is sufficiently large, and if protection from trade allows it to catch up, then temporary autarky can increase overall consumer surplus for the high cost country.

Proof: Result 2, Appendix A.

Corollary 2. If the high cost firm's distance to frontier is sufficiently large, and if protection from trade allows it to catch up, then temporary autarky can increase overall consumer surplus for the low cost country.

Proof: Result 3, Appendix A.¹²

^{12.} Note that corollaries 1 and 2 imply that even in the absence of a public good, it may be welfare-improving to temporarily refrain from trade to allow an infant industry to catch up. While this is not a new insight from the perspective of the country where the infant industry is located, see e.g. Melitz 2005; Young 1991, corollary 2 further implies that consumers in the initially frontier country may also benefit.

Remark 1. If public goods provision is higher under temporary autarky than under free trade, then consumer surplus in both countries is also higher under temporary autarky. The reverse is not the case. Further, the high cost country makes a greater temporary sacrifice in terms of consumer surplus by choosing autarky.

Figure 1a plots the ratio of each country's overall (period 1 plus period 2) consumer surplus, as well as overall consumption and thus public goods provision, under trade versus autarky for different values of d and b when a = 2c.

Proposition 2. If the high cost firm is close enough to the technology frontier to present sufficient competitive pressure under trade, then the low cost firm's profits are higher under temporary protection than free trade. However, there are no values of d under which both consumer surplus and public goods provision and profits in the low cost country are higher under protection.

Proof: Result 4, Appendix A.

Figure 1b plots the ratio of profits made by firm *i* under trade over those made under temporary autarky, illustrating result 4. Figure 1c plots the ratios of country welfare. It shows that for b = 0, losses in rents and period 1 consumer surplus outweigh any period 2 gains in consumer surplus from the infant industry catching up for the low cost country, while the threshold for infant industry protection being beneficial is much lower for the high cost country. The greater the positive externality *b*, the more closely aligned countries' preferences for trade or temporary autarky become. Finally, Figure 1d plots the ratios of global welfare for different values of *b*, showing that in the absence of a public good, global welfare implications more closely resemble the low cost country's welfare and make free trade the preferred choice. When a positive externality is present, infant industry protection becomes welfare improving globally for sufficiently high values of *d*.

5 TRADE AND AUTARKY WITH SUBSIDIES

The previous section has examined the conditions under which countries will find it beneficial to move from autarky to trade from period 1, assuming that no other policy levers are available. However, given the presence of two market failures (the positive consumption externality and imperfect competition), governments may wish to employ corrective measures, such as subsidies.

In principle, the government can subsidize either production or consumption of the technology. An upstream subsidy shifts the firm's profit

$$\Pi_i = (P(q_i) + s_i^{producer})q_i - cq_i \tag{18}$$



(c) Country Welfare

(d) Global Welfare

Note: The figures above plot the ratios of overall (sum of period 1 and period 2) consumer surplus, consumption of the good, firm *i*'s profit, country welfare, and global welfare for different values of *d* under trade as compared to autarky when a = 2c. Figures 5, 6 and 7 (Appendix) show the same plots for different ratios of *a* to *c*. Where the ratio falls below 1, the outcome in question is improved by remaining in atuarky in period 1.

The figures illustrate the opposing effects of trade on profit and consumer surplus: the further country *j* is from the technological frontier, the more trade reduces overall consumer surplus, and the more it increases firm *i*'s profits (firm *j*'s profits are not plotted as the ratio is always 0). Figures 1c and 1d highlight the significance of the positive externality b * R in the model: for b = 0, trade is always welfare improving for the low-cost country, as well as globally, for any value of *d* which is greater than 1 and satisfies equation 17. However, for positive values of *b*, trading in both periods can become welfare reducing beyond certain thresholds of *d* for both countries. Moreover, the greater *b* becomes relative to *a*, the more individual countries' preferences should resemble each other.

FIGURE 1

Consumer Surplus, Consumption, Profit and Welfare Under Trade Relative to Autarky

while a downstream subsidy shifts the consumer's utility from consuming the technology (and thus demand)

$$U_i(r_i) = u(r_i) - (P - s_i^{consumer})r_i$$
⁽¹⁹⁾

Under autarky, it makes no difference which subsidy is used. Under trade, this may no longer be the case.¹³

5.1 Consumer Subsidies

In each stage, under autarky each government will subsidize in order to eliminate the deadweight loss from the monopoly and internalize the domestic part of the climate externality. Consider a downstream subsidy in country i. A quantity subsidy s shifts consumer demand such that

$$\frac{\delta u}{\delta r_i} + s_i^{consumer} = P \tag{20}$$

The monopolist maximizes

$$\Pi_{i} = \left(\frac{\delta u}{\delta r_{i}} + s_{i}^{consumer}\right)r_{i} - cr_{i}$$

$$\rightarrow \frac{\delta^{2} u}{\delta r_{i}}r_{i} + \frac{\delta u}{\delta r_{i}} + s_{i}^{consumer} = c$$
(21)

The government will wish to subsidize such that $\frac{\delta u}{\delta r_i} + \frac{b}{2} = c$. With $u(r_i) = ar_i - \frac{r_i^2}{2}$, we get

$$s_i = a + b - c; r_i = a - c + \frac{b}{2}$$
 (22)

Proof: Appendix B.1. The high cost country solves an equivalent problem, substituting dc for c.

Under trade, the low cost firm once again supplies the world market at P = dc. The low cost country's government will once again want to set a subsidy such that $\frac{\delta u}{\delta r_i} + \frac{b}{2} = c$. Since $\frac{\delta u}{\delta r_i} = P - s_i^{consumer} = dc - s_i^{consumer}$, $c - \frac{b}{2} = dc - s_i^{consumer}$ and we obtain

$$s_i^{consumer} = c(1-d) + \frac{b}{2}$$
(23)

The high cost firm is inactive and makes no profit, implying that prices and marginal costs of consumption in the high cost country are equal at dc. The high cost country thus sets a subsidy such that $\frac{\delta u}{\delta r_j} + \frac{b}{2} = dc$, which given that $\frac{\delta u}{\delta r_j} = P - s_j^{consumer} = dc - s_j^{consumer}$, implies

^{13.} Fischer et al. 2017 highlight how under imperfect competition, an upstream subsidy will enhance the domestic firm's market share, while a downstream subsidy benefits both domestic and foreign firms. Under Cournot competition, a downstream subsidy also tends to increase prices globally, which is not the case here.

$$s_j^{consumer} = \frac{b}{2} \tag{24}$$

The low cost country extracts rent amounting to

$$Rent_i = c(1-d)r_i \tag{25}$$

Proposition 3. Under Bertrand competition and with downstream subsidies, the low cost country chooses a subsidy which internalizes the domestic part of the public good and eliminates the domestic loss in consumer surplus from the monopoly, which is proportionate to the rent which the low cost firm extracts due to the high cost firm's distance to frontier. On net, the low cost country thus extracts rent from the high cost country's consumers. The high cost country, whose firm is inactive, sets a subsidy which only internalizes the domestic externality. Quantities, consumer surplus and public goods provision are the same under trade as under autarky.

Proof: Result 5, Appendix B.1.

Given Result 5 and equation 1, it is trivial to show that autarky in Stage 1 leads to higher public goods provision and higher welfare in the high cost country in Stage 2 for all parameter values.¹⁴ The low cost country may yet prefer free trade if the rents it can thereby collect from the high cost country outweigh the loss in public goods provision.

Proposition 4. Under Bertrand competition and downstream subsidies, Stage 1 autarky leads to an increase in public goods provision and consumer surplus over both periods for both countries. This is a pareto improvement from the low cost country's perspective if these gains are not outweighed by the loss in rents collected from the high cost country.

This implies that when infant industry protection allows an initially uncompetitive industry to catch up, *and* the new technology is at a stage where competitive pressures on first mover industries are low, then protection is always preferable from the point of view of the country where the infant industry is located; it increases climate action globally; and it may be a pareto improvement in some instances.

5.2 **Producer Subsidies**

Under autarky, it makes no difference whether countries set an upstream or downstream subsidy. Equilibrium outcomes at any stage of the game under autarky are therefore identical to those derived in section 5.1. Under trade, producer subsidies affect the prices faced by both countries' consumers, as well as the market share of the domestic firm. In Stage 1, the industry is unlevelled. In Stage 2, it may be levelled (if the low cost country caught up, either by

^{14.} The Stage 1 outcome from the perspective of the high cost country is identical under trade or autarky. In Stage 2, if countries were previously in autarky, both firms produce at marginal cost c, leading to higher consumption and higher public goods provision in the high cost country in Stage 2.

remaining in autarky or by trading with a sufficient production subsidy) or unlevelled (if countries trade in Stage 1 and the low cost firm supplies the whole market, or the high cost country subsidized its firm sufficiently to price the low cost firm out of the market).

Levelled Industry Suppose we start with a level playing field in Stage 2, such that both firms have constant marginal cost equal to *c*. If both firms share the market, then price equals $c - s_i = c - s_j$ and each government solves

$$\max_{\{s_i\}} W_i(s_i, s_j) = \max_{\{s_i\}} CS(s_i, s_j) + \Pi(s_i, s_j) + \frac{b}{2}R(s_i, s_j) - s_i \frac{R(s_i, s_j)}{2}$$
(26)

Note that under Bertrand competition and marginal cost pricing, profits equal 0 and the government weighs consumer surplus and public goods provision against the cost of the subsidy. The first order condition yields $s_i = s_j = b$ – the global optimum. Is this a Nash Equilibrium? Appendix B.2 shows that it is not. Under Bertrand competition and producer subsidies, no rents can be extracted from the importing country to compensate for the cost of the subsidy. Each country will therefore wish to reduce subsidies, thereby shifting the burden of public goods provision onto the other country, up until the point at which the subsidy is optimal even if the country is the sole supplier and subsidizer of the technology.

Proposition 5. In a levelled industry, if countries trade they will set identical producer subsidies. Because a producer subsidy subsidizes consumer surplus and the public good in the other country as well as domestically, and no rents can be extracted, neither country has an incentive to price the other out of the market. Producer subsidies, consumption and public goods provision are lower under producer than under consumer subsidies.

Proof: Result 6, Appendix B.2.

Unlevelled Industry Suppose country *i* has marginal cost *c* and country *j* has marginal cost *dc*. Market sharing requires a subsidy on the part of country *j* which it will not be prepared to support, unless it does not expect to pay it. Proof: Appendix B.2. The low cost firm therefore supplies the whole market, while the high cost government sets a producer subsidy which enforces marginal cost pricing, but does not lead to actual production in the high cost country.

Since the low cost government subsidizes both markets, while both countries receive the same consumer surplus and public goods provision, the high cost country's welfare is higher in this scenario.¹⁵ What does this imply for Stage 1?

Subgame Perfect Nash Equilibrium Since it is not desirable to be in the position of the low cost country in Stage 2, if countries trade then the high cost country will not be prepared to

^{15.} Note that I am abstracting from other potential benefits of economic activity, such as employment. Moreover, the very stylized nature of Bertrand competition and associated profit dissipation is a key assumption here.

subsidize consumption enough to participate in global trade.¹⁶ It is therefore in the low cost country's interest not to trade in Stage 1, if the benefits of not having to subsidize the high cost country in Stage 2 outweigh the costs of lower public goods provision in Stage 1. Moreover, it is not obvious that public goods provision is greater under trade. This would require that $\frac{2}{3}(2a-2c+b) > 2a-c(1+d)+b$, which holds only for very large values of d.¹⁷

Furthermore, after Stage 1 autarky it is in both countries' interests to remain in autarky in Stage 2, since public goods provision is higher under autarky than trade for d = 1. Overall, upstream subsidies to provide the public good are not desirable, as they increase opportunities to free-ride and reduce subsidies and consumption in equilibrium as compared to downstream subsidies. If downstream subsidies are available, neither country will choose upstream subsidies to support the public good.

Proposition 6. Results 6 and 7 imply that regardless of whether the industry is levelled or not, producer subsidies result in lower quantities and public goods provision than consumer subsidies, as they incentivize free riding. In an unlevelled industry, only the low cost firm is active, and the low cost country bears the full cost of public goods provision.

Temporary Production Subsidy Can an upstream subsidy ever be welfare improving? The low cost country will never choose a production over a consumption subsidy if given the choice, as much of the subsidy would 'leak out' or 'spill over' to the other country. Paragraph B.2 considers the case in which countries trade and the high cost country subsidizes its industry during Stage 1 in order to catch up with the low cost industry. A temporary production subsidy is preferable to autarky if the gains in consumer surplus and the public good outweigh the cost of the subsidy.

Proposition 7. In a 2-stage game with learning according to equation 1, the high cost country always prefers trade with a temporary production subsidy to trade without it. Whether a production subsidy is preferred to autarky depends on whether the associated gains in consumer surplus and the public good outweigh the cost of the subsidy.

However, if the high cost country can choose its subsidy mix in both periods, then under Bertrand competition it will find it preferable not to actually catch up, but to set a subsidy which enforces marginal cost pricing on the part of the low cost firm.

5.3 Equilibrium Policy Mix

If both countries choose their subsidy mix in both periods, then under trade, the high cost country will set an upstream subsidy which enforces marginal cost pricing, ie $s_i^{producer} = c(d - c)$

^{16.} Table 3 shows the pay-off matrix for the low cost country.

^{17.} So large that it would not be consistent with our previous condition that $dc < \frac{a+c}{2}$, and is consistent with a > dc only for values of a so large that a - c > b.

1) – ε , while both countries will set a downstream subsidy which equals $s_i^{consumer} = s_j^{consumer} = \frac{b}{2}$. The low cost firm supplies the world market at P = c, with $r_i = r_j = a - c$. Welfare in each country during each stage of the game equals

$$W_i = W_j = \frac{(a-c)^2}{2} + b(a-c)$$
(27)

This is the Subgame Perfect Nash Equilibrium of this game.

To see this, note that the high cost country cannot improve its welfare by deviating from this position. If it sets a higher upstream subsidy, both firms will either share the market, which reduces W_j by $c(d-1)r_j$, or the high cost firm supplies the whole market, which reduces W_j by c(d-1)R. Given the price regime and trade conditions, $\frac{b}{2}$ is the optimal consumer subsidy. While public goods provision would increase if the subsidy was higher, each country considers only the marginal benefit to its own population, which equals $\frac{b}{2}$. Further, since the high cost country does not actually incur the upstream subsidy, it has no incentive to move to autarky, as this would increase production costs and reduce consumer surplus and public goods provision.

The low cost country cannot improve welfare by changing the subsidy regime either, for the same reasons. When production subsidies are available, it is unable to extract rent from the high cost country in either period. In autarky, the low cost country would set a subsidy which eliminates the deadweight loss from the monopoly and internalizes the domestic part of the public good, leading to the same domestic outcome as that which is obtained when $s_j^{producer} = c(d-1) - \varepsilon$ and $s_i^{consumer} = s_j^{consumer} = \frac{b}{2}$. Autarky would reduce welfare in the low cost country by lowering public goods provision in the high cost country.

The policy mix countries will choose if they are able to trade and use a combination of producer and consumer subsidies thus delivers the highest cumulative welfare, given the first best is not available. It cannot correct the market failure resulting from the climate externality. However, it yields an outcome equivalent to perfect competition. On the other hand, if producer subsidies are not available, trade may be welfare reducing, as sections 4, 5.1 and 5.2 have shown.

6 CASE STUDY: SOLAR PHOTOVOLTAICS

Renewable energy technologies, once considered too expensive to be economically viable, have seen dramatic declines in cost over the past few decades, becoming competitive with traditional fossil fuels in many contexts. The two great success stories are electricity production using wind and solar power. Since the first commercial use of solar PV in 1958, its cost decreased by more than three orders of magnitude (Way et al. 2022). The price of solar panels fell by 75% between 2010 and 2015 (Gerarden 2017).

While about half of these cost declines can be attributed to reductions in material costs, economies of scale, and efficiency-increasing innovation (Nemet 2006), increased competition



is thought to be another key driver (Carvalho et al. 2017; Dent 2018; Nemet 2006).

Note: The green line plots the Levelized Cost of Electricity from solar PV in \$ per kWh between 2000 and 2016. The blue bars show global electricity generation from solar PV in TWh over the same period. We see cost declining from over 0.5 to 0.1 \$ per kWh, while deployment rose from close to zero to over 70 TWh. Source: Author's calculations based on Way et al. 2022 and Dudley et al. 2018.

Figure 2 shows the levelized cost of electricity¹⁸ in solar PV, as well as global electricity generation from solar, over the period 2000-2016. During this time, the levelized cost of electricity declined from over 0.5 to 0.1 USD per kWh, while global electricity generation increased from virtually nothing to more then 70 TWh per year. Technology innovation support and demand creation through government subsidies around the world, as well as the expansion of Chinese manufacturing – characterized by both supply- and demand-side industrial policy (cf. Chen 2015) – are likely to have played a significant role.¹⁹ Early government policies supporting solar PV include, for example, government R&D support and the founding of the Solar Energy Research Institute in the US in 1974 and the 1990 '1000 Roof' solar deployment program, as well as the 1991 solar feed-in tariff, in Germany (Hansen et al. 2018).

Figure 3 illustrates how the share of the top 10 firms in terms of global solar PV shipments declined from 88% in 2000 to 53% in 2015^{20} – a period which experienced significant firm entry, in particular by Chinese firms. The Chinese government supported its solar PV

^{18.} The price per unit of electricity which would be required in order for a project to break even over its lifetime

^{19.} In the context of the model presented in this paper, measures targeting the supply side (such as R&D support or export subsidies) would be conceptualized as 'producer subsidies', while demand-creation policies (such as feed-in tariffs or other deployment programmes) are conceptualized as 'consumer subsidies'.

^{20.} Note that the data on market concentration displayed in this section was sourced from industry blogs, as it was not possible to obtain data going back further than 2015 and covering more than the top 5 players from more formal data providers.



Note: The Figure shows the share of the top 10 producers in global shipments of solar PV generation capacity for the years 2000 (when the top 10 captured 88% of global market share), 2005, 2010, and 2015 (when the share of the top 10 had declined to 53%). Source: Author's calculations based on Mints 2016; Renewable Energy World 2014.



industry using a mix of upstream subsidies, including discounts on raw materials, electricity and funding, export subsidies, and technological, infrastructure and personnel support (Chen 2015); and demand-creation measures such as feed-in tariffs and free grid connection services for distributed solar by China's largest state-owned utility (Zhang et al. 2013). Chinese solar panel manufacturers reached more than 50% of global revenue share by 2012 (Chen 2015).

In 2012, the US and the EU both imposed anti-dumping duties on Chinese solar panels, arguing that the latter were unfairly subsidized (Hughes et al. 2017; Meckling et al. 2018) and thereby retaliating against subsidies which reduced the cost of a low-carbon energy technology for their own utilities and consumers. The model presented here provides intuition for why the loss in profits earned by domestic producers may have outweighed the benefits to consumers and the climate from the perspective of western governments.

Looking forward, technologies which need to decline in cost in order to be economical include carbon capture and storage, green hydrogen, and energy storage. Lithium ion batteries, for example, are currently the most expensive part of electric vehicles, with a heavily concentrated global market. Figure 4 plots global market shares in EV batteries in 2022, showing that the top 10 producers currently capture 92% of global market share. Firm entry and competition may be key to driving down cost.

However, it is important to note that competition is only one of many reasons why the solar sector has evolved as it has. Moreover, the ability of Chinese manufacturers to sell at

lower prices is likely to be at least in part due to domestic production conditions, rather than rent-seeking by western manufacturers.²¹ Finally, the relationship between firm entry and technological maturity is surely bi-directional. While this section has sought to add intuition and real-world context to the theoretical findings, it should be interpreted with caution.





FIGURE 4 Market Concentration in 2022 (EV Batteries)

7 CONCLUSION: IMPLICATIONS FOR TRADE POLICY

This paper has used a simple, two-country model of trade in a technology with positive externalities to highlight how in the absence of subsidies, temporary autarky to allow an infant industry to catch up may increase consumer surplus and public goods provision and be welfare improving for the high cost country, as well as potentially the low cost country, if the high cost country's distance to frontier is sufficiently large. This is driven by the losses otherwise incurred due to imperfect competition. The low cost country may prefer that the high cost country does not catch up, if the rents it thereby extracts outweigh gains in public goods provision and consumer surplus.

^{21.} One seemingly obvious advantage are lower labor costs. However, Chen 2015 argues that labor costs account for only 10% of the cost of solar panel manufacturing, due to the industry's highly capital intensive nature.

When both countries use consumer subsidies which correct monopoly losses as well as internalizing the domestic public good, public goods provision and consumer surplus under temporary autarky are higher for any distance to frontier – whether the low cost country also prefers autarky depends on whether this outweighs the loss in rent extracted. When both countries use producer subsidies, autarky in both periods is preferable to the low cost country, because producer subsidies in a set-up with Bertrand competition and profit dissipation exacerbate free riding on the other country's subsidies, leading to lower public goods provision and lower welfare for the supplier country.

When countries are free to choose their subsidy mix, the high cost country can subsidize its domestic industry to enforce marginal cost pricing under trade. This is the Sub-Game Perfect Nash Equilibrium of this game. The low cost country may find that producer subsidies lower its domestic welfare due to the loss in rents extracted from the high cost country. Put differently: allowing the laggard country to more effectively compete with the frontier country through upstream subsidies²² can increase consumer surplus and public goods provision globally. However, it is at odds with any mercantilist ambitions to extract rents which the initially frontier country may harbor.

The model provides intuition for why it may be optimal that (a) countries supporting clean technology often use trade barriers such as local content requirements, to the extent to which those actually work,²³ and (b) early movers in the global market often oppose production subsidies, as was apparent in the EU and US-China solar trade wars, for example. The results imply that an environmental trade agreement may be desirable from a climate point of view only when production subsidies are available. Global trade law does not currently make allowances for the potential global benefits of producer subsidies for products with positive externalities, which renders such subsidies susceptible to challenge, as exemplified in the US- and EU-China solar trade wars. This relates to the broader challenge of reviewing WTO rules to ensure they are compatible with climate goals, especially given stated plans to introduce a Carbon Border Adjustment Mechanism in the European Union (Grubb et al. 2022). More broadly, encouraging the use of producer support over protectionist measures accompanying demand-side policies could reduce the costs of protectionism and may prove more effective than simply challenging local content requirements or discriminatory public procurement.

There are a number of important limitations. First, Bertrand competition abstracts from some of the potential benefits of having a competitive industry in a level playing field, due to rent dissipation. Further work on the trade-offs between trade and autarky may investigate how Cournot competition might lead to different outcomes by providing stronger incentives to compete for market share. However, with a large number of entrants following the maturation of infant industries I would argue that it is reasonable to assume profit dissipation.

^{22.} And for some parameter values, even infant industry protection, which is more costly for both countries.

^{23.} Recent empirical evidence, however, suggests that local content requirements alone are not sufficient to develop an infant industry, see Scheifele et al. 2022.

Moreover, I use a very simple model of learning in which the laggard country catches up immediately if it is active, and the frontier country does not learn. Future work could consider a more sophisticated learning curve, as well as dynamics of innovation in level versus uneven global supplier markets. The merits of infant industry protection would then depend on the relative rates of learning, as well as degree to which current market share and competition incentivize innovation. Finally, the paper only considers infant industry protection and the competitive implications of a level versus an un-level playing field. Future work could look at the implications of 'racing' to be the frontier country and extract rents in the first place, which may provide a channel through which to address the global collective action problem presented by climate change.

REFERENCES

- Aghion, Philippe, Antonin Bergeaud, Matthieu Lequien, Marc J Melitz, et al. 2018. "The Impact of Exports on Innovation: Theory and Evidence." *NBER Working Paper No.* 24600.
- Andreoni, Antonio, and Ha-Joon Chang. 2016. "Industrial Policy and the Future of Manufacturing." *Economia E Politica Industriale* 43 (4): 491–502.
- Baldwin, John R, and Wulong Gu. 2004. "Trade Liberalization: Export-market Participation, Productivity Growth, and Innovation." *Oxford Review of Economic Policy* 20 (3): 372–392.
- Balineau, Gaëlle, and Jaime De Melo. 2013. "Removing Barriers to Trade on Environmental Goods: An Appraisal." *World Trade Review* 12 (4): 693–718.
- Barrett, Scott. 1994a. "Self-enforcing International Environmental Agreements." Oxford Economic Papers 46:878–94.
 - ——. 1994b. "Strategic Environmental Policy and International Trade." *Journal of Public Economics* 54 (3): 325–338.
- ———. 1997. "The Strategy of Trade Sanctions in International Environmental Agreements." *Resource and Energy Economics* 19 (4): 345–361.
- ———. 2005. "The Theory of International Environmental Agreements." *Handbook of Environmental Economics* 3:1457–1516.

. 2006. "Climate Treaties And 'Breakthrough' Technologies." American Economic Review 96 (2): 22–25.

- Barrett, Scott, and Astrid Dannenberg. 2022. "The Decision to Link Trade Agreements to the Supply of Global Public Goods." *Journal of the Association of Environmental and Resource Economists* 9 (2): 273–305.
- Battaglini, Marco, and Bård Harstad. 2016. "Participation and Duration of Environmental Agreements." *Journal of Political Economy* 124 (1): 160–204.
- Beccherle, Julien, and Jean Tirole. 2011. "Regional Initiatives and the Cost of Delaying Binding Climate Change Agreements." *Journal of Public Economics* 95 (11-12): 1339–1348.
- Brito, Dagobert L, Eytan Sheshinski, and Michael D Intriligator. 1991. "Externalities and Compulsary Vaccinations." *Journal of Public Economics* 45 (1): 69–90.
- Carbaugh, Robert J, and Max St Brown. 2012. "Industrial Policy and Renewable Energy: Trade Conflicts." *Journal of International and Global Economic Studies*.
- Carvalho, Maria, Antoine Dechezleprêtre, Matthieu Glachant, et al. 2017. "Understanding the Dynamics of Global Value Chains for Solar Photovoltaic Technologies." *WIPO Economic Research Working Paper No. 40* 40.
- Chang, Ha-Joon. 2003. "Kicking Away the Ladder: Infant Industry Promotion in Historical Perspective." *Oxford Development Studies* 31 (1): 21–32.
- Chen, Gang. 2015. "From Mercantile Strategy to Domestic Demand Stimulation: Changes in China's Solar PV Subsidies." *Asia Pacific Business Review* 21 (1): 96–112.

- De Melo, Jaime, and Jean-Marc Solleder. 2020. "Barriers to Trade in Environmental Goods: How Important They Are and What Should Developing Countries Expect from Their Removal." World Development 130:104910.
 - —. 2022. "Towards an Environmental Goods Agreement Style (EGAST) Agenda to Improve the Regime Complex for Climate Change." In *Handbook on Trade Policy and Climate Change*, 202–219. Edward Elgar Publishing.
- Dent, Christopher M. 2018. "Clean Energy Trade Governance: Reconciling Trade Liberalism and Climate Interventionism?" *New Political Economy* 23 (6): 728–747.
- Droege, Susanne, Harro Van Asselt, Kasturi Das, and Michael Mehling. 2016. "The Trade System and Climate Action: Ways Forward under the Paris Agreement." *Scj Int'l L. & Bus.* 13:195.
- Dudley, Bob, et al. 2018. "BP Statistical Review of World Energy." BP Statistical Review, London, UK 6:2018.
- E-Vehicle Info. 2022. "Top 10 EV Battery Manufacturers in World by Market Share." Accessed 6 December 2022, https://e-vehicleinfo.com/global/ev-battery-manufacturers-in-world-by-market-share/.
- Fajgelbaum, Pablo D, Pinelopi K Goldberg, Patrick J Kennedy, and Amit K Khandelwal. 2020. "The Return to Protectionism." *The Quarterly Journal of Economics* 135 (1): 1–55.
- Finus, Michael. 2008. "Game Theoretic Research on the Design of International Environmental Agreements: Insights, Critical Remarks, and Future Challenges." *International Review of Environmental and Resource Economics* 2 (1): 29–67.
- Fischer, Carolyn, Mads Greaker, and Knut Einar Rosendahl. 2017. "Robust Technology Policy against Emission Leakage: The Case of Upstream Subsidies." *Journal of Environmental Economics and Management* 84:44–61.
 - ——. 2018. "Strategic Technology Policy As a Supplement to Renewable Energy Standards." *Resource and Energy Economics* 51:84–98.
- Fisman, David N, and Kevin B Laupland. 2009. "The Sounds of Silence: Public Goods, Externalities, and the Value of Infectious Disease Control Programs." *Canadian Journal of Infectious Diseases and Medical Microbiology* 20 (2): 39–41.
- Francis, Peter J. 1997. "Dynamic Epidemiology and the Market for Vaccinations." *Journal of Public Economics* 63 (3): 383–406.
- Gerarden, Todd. 2017. "Demanding Innovation: The Impact of Consumer Subsidies on Solar Panel Production Costs." *Harvard Environmental Economics Program*.
- Grossman, Gene M, and Elhanan Helpman. 1990. "Trade, Innovation, and Growth." *The American Economic Review* 80 (2): 86–91.
- Grubb, Michael, Nino David Jordan, Edgar Hertwich, Karsten Neuhoff, Kasturi Das, Kaushik Ranjan Bandyopadhyay, Harro Van Asselt, Misato Sato, Ranran Wang, William A Pizer, et al. 2022. "Carbon Leakage, Consumption, and Trade." *Annual Review of Environment* and Resources 47:753–795.

- Hagen, Achim, and Jan Schneider. 2021. "Trade Sanctions and the Stability of Climate Coalitions." Journal of Environmental Economics and Management 109:102504.
- Hanlon, William. 2017. "Dynamic Comparative Advantage in International Shipbuilding: The Transition from Wood to Steel." In 2017 Meeting Papers. 140. Society for Economic Dynamics.
- Hansen, Erik G, Florian Lüdeke-Freund, Xiaohong Iris Quan, and Joel West. 2018. "Cross-National Complementarity of Technology Push, Demand Pull, and Manufacturing Push Policies: The Case of Photovoltaics." *IEEE Transactions on Engineering Management* 66 (3): 381–397.
- Harstad, Bård. 2012. "Climate Contracts: A Game of Emissions, Investments, Negotiations, and Renegotiations." *Review of Economic Studies* 79 (4): 1527–1557.
- ——. 2016. "The Dynamics of Climate Agreements." *Journal of the European Economic Association* 14 (3): 719–752.
- Hausmann, Ricardo, Jason Hwang, and Dani Rodrik. 2007. "What You Export Matters." *Journal of Economic Growth* 12 (1): 1–25.
- Helm, Dieter, Cameron Hepburn, and Giovanni Ruta. 2012. "Trade, Climate Change, and the Political Game Theory of Border Carbon Adjustments." Oxford Review of Economic Policy 28 (2): 368–394.
- Hoel, Michael, and Aart De Zeeuw. 2010. "Can a Focus on Breakthrough Technologies Improve the Performance of International Environmental Agreements?" *Environmental and Resource Economics* 47 (3): 395–406.
- Houde, Sebastien, and Wenjun Wang. 2022. "The Incidence of the US-China Solar Trade War." *Economics Working Paper Series No. 22/372.*
- Hughes, Llewelyn, and Jonas Meckling. 2017. "The Politics of Renewable Energy Trade: The US-China Solar Dispute." *Energy Policy* 105:256–262.
- ICTSD. 2011. Fostering Low Carbon Growth: The Case for a Sustainable Energy Trade Agreement. International Centre for Trade / Sustainable Development.
- IEA. 2022. Solar PV Global Supply Chains. Technical report. International Energy Agency.
- Irwin, Douglas A. 2000. "Did Late-nineteenth-century US Tariffs Promote Infant Industries? Evidence from the Tinplate Industry." *The Journal of Economic History* 60 (2): 335–360.
- Jacob, Arun, and Anders K Møller. 2017. *Policy Landscape of Trade in Environmental Goods and Services*. Technical report. Artnet Working Paper Series.
- Johnson, Oliver. 2016. "Promoting Green Industrial Development through Local Content Requirements: India's National Solar Mission." *Climate Policy* 16 (2): 178–195.
- Juhász, Réka. 2018. "Temporary Protection and Technology Adoption: Evidence from the Napoleonic Blockade." *American Economic Review* 108 (11): 3339–76.
- Keller, Wolfgang. 2004. "International Technology Diffusion." *Journal of Economic Literature* 42 (3): 752–782.

- Krueger, Anne O, and Baran Tuncer. 1982. "An Empirical Test of the Infant Industry Argument." *The American Economic Review* 72 (5): 1142–1152.
- Krugman, Paul R. 1979. "Increasing Returns, Monopolistic Competition, and International Trade." *Journal of International Economics* 9 (4): 469–479.
- Markusen, James R. 1981. "Trade and the Gains From Trade With Imperfect Competition." *Journal of International Economics* 11 (4): 531–551.
- Meckling, Jonas, and Llewelyn Hughes. 2018. "Protecting Solar: Global Supply Chains and Business Power." *New Political Economy*, 88–104.
- Melitz, Marc J. 2003. "The Impact of Trade on Intra-industry Reallocations and Aggregate Industry Productivity." *Econometrica* 71 (6): 1695–1725.
- ——. 2005. "When and How Should Infant Industries Be Protected?" *Journal of International Economics* 66 (1): 177–196.
- Mints, Paula. 2016. "2015 Top Ten PV Cell Manufacturers." Accessed 6 December 2022, https://www.renewableenergyworld.com/solar/2015-top-ten-pv-cell-manufacturers/#gref.
- Monkelbaan, Joachim. 2017. "Using Trade for Achieving the SDGs: The Example of the Environmental Goods Agreement." *Journal of World Trade* 51 (4).
- Nemet, Gregory F. 2006. "Beyond the Learning Curve: Factors Influencing Cost Reductions in Photovoltaics." *Energy Policy* 34 (17): 3218–3232.
- ——. 2019. "Chinese Entrepreneurs." In How Solar Energy Became Cheap, 133–158. Routledge.
- Nordhaus, William. 2015. "Climate Clubs: Overcoming Free-riding in International Climate Policy." *American Economic Review* 105 (4): 1339–70.
- Renewable Energy World. 2014. "Top Ten PV Manufacturers from 2000 to Present: A Pictorial Retrospective." Accessed 6 December, 2022, https://www.renewableenergyworld.com/solar/top-ten-pv-manufacturers-from-2000-to-present-a-pictorial-retrospective/#gref.
- Ricardo, David. 1891. Principles of Political Economy and Taxation. G. Bell / Sons.
- Richter, Philipp M, Marco Runkel, and Robert C Schmidt. 2021. "Strategic Environmental Policy and the Mobility of Firms." *Journal of the Association of Environmental and Resource Economists* 8 (5): 863–893.
- Scheifele, Fabian, Moritz Bräuning, and Benedict Probst. 2022. "The Impact of Local Content Requirements on the Development of Export Competitiveness in Solar and Wind Technologies." *Renewable and Sustainable Energy Reviews* 168:112831.
- Steenblik, Ronald. 2005. *Environmental goods: A comparison of the APEC and OECD lists.* Technical report. OECD Publishing.
- Traiberman, Sharon, and Martin Rotemberg. 2022. "Precautionary Protectionism." NBER Working Paper No. 30300.
- Vossenaar, Rene. 2016. "Reducing Import Tariffs for Environmental Goods: The APEC Experience." *International Centre for Trade and Sustainable Development Issue Paper No.* 22.

- Way, Rupert, Matthew C Ives, Penny Mealy, and J Doyne Farmer. 2022. "Empirically Grounded Technology Forecasts and the Energy Transition." *Joule* 6 (9): 2057–2082.
- Wu, Mark, and James Salzman. 2013. "The Next Generation of Trade and Environment Conflicts: The Rise of Green Industrial Policy." *Nw. Ul Rev.* 108:401.
- Young, Alwyn. 1991. "Learning by Doing and the Dynamic Effects of International Trade." *The Quarterly Journal of Economics* 106 (2): 369–405.
- Zhang, Sufang, and Yongxiu He. 2013. "Analysis on the Development and Policy of Solar PV Power in China." *Renewable and Sustainable Energy Reviews* 21:393–401.

Appendix

A TRADE AND AUTARKY WITHOUT SUBSIDIES

Stage 1 Outcome, country i	Stage 2 Outcome, country <i>i</i>
$P_i = \frac{a+c}{2}; r_i = \frac{a-c}{2}; R = a - \frac{c(1+d)}{2}$	$P_i = c; r_i = a - c; R = 2(a - c)$
$W_i = \frac{3}{8}(a-c)^2 + \frac{b}{2}(a - \frac{c(1+d)}{2})$	$W_i = \frac{(a-c)^2}{2} + b(a-c)$
$CS_i = \frac{(a-c)^2}{8}; \Pi_i = \frac{(a-c)^2}{4}$	$CS_i = rac{(a-c)^2}{2}; \Pi_i = 0$
$P_i = dc; r_i = a - dc; R = 2(a - dc)$	$P_i = dc; r_i = a - dc; R = 2(a - dc)$
$W_i = (a - dc) \left(\frac{a - dc}{2} + 2c(d - 1) + b \right)$	$W_i = (a - dc) \left(\frac{a - dc}{2} + 2c(d - 1) + b\right)$
$CS_i = \frac{(a-dc)^2}{2}; \Pi_i = 2c(a-dc)(d-1)$	$CS_i = \frac{(a-dc)^2}{2}; \Pi_i = 2c(a-dc)(d-1)$
	Stage 1 Outcome, country <i>i</i> $P_{i} = \frac{a+c}{2}; r_{i} = \frac{a-c}{2}; R = a - \frac{c(1+d)}{2}$ $W_{i} = \frac{3}{8}(a-c)^{2} + \frac{b}{2}(a - \frac{c(1+d)}{2})$ $CS_{i} = \frac{(a-c)^{2}}{8}; \Pi_{i} = \frac{(a-c)^{2}}{4}$ $P_{i} = dc; r_{i} = a - dc; R = 2(a - dc)$ $W_{i} = (a - dc) \left(\frac{a-dc}{2} + 2c(d-1) + b\right)$ $CS_{i} = \frac{(a-dc)^{2}}{2}; \Pi_{i} = 2c(a - dc)(d-1)$

TABLE 1Outcome Matrix for Country i

Note: The Table shows prices, quantities, profits, consumer surplus, overall public goods provision and welfare for the low cost country in both Stages of the game, depending on whether countries are in autarky or engage in trade during Stage 1.

TABLE 2Outcome Matrix for Country j

Stage 1 Strategy	Stage 1 Outcome, country j	Stage 2 Outcome, country j
	$P_j = \frac{a+dc}{2}; r_j = \frac{a-dc}{2}; R = a - \frac{c(1+d)}{2}$	$P_j = c; r_j = a - c; R = 2(a - c)$
Autarky	$W_j = \frac{3}{8}(a - dc)^2 + \frac{b}{2}(a - \frac{c(1+d)}{2})$	$W_j = \frac{(a-c)^2}{2} + b(a-c)$
	$CS_j = \frac{(a-dc)^2}{8}; \Pi_j = \frac{(a-dc)^2}{4}$	$CS_j = \frac{(a-c)^2}{2}; \Pi_j = 0$
	$P_j = dc; r_j = a - dc; R = 2(a - dc)$	$P_j = dc; r_j = a - dc; R = 2(a - dc)$
Trade	$W_j = \frac{(a-dc)^2}{2} + b(a-dc)$	$W_j = \frac{(a-dc)^2}{2} + b(a-dc)$
	$CS_j = rac{(a-dc)^2}{2}; \Pi_j = 0$	$CS_j = \frac{(a-dc)^2}{2}; \Pi_j = 0$

Note: The Table shows prices, quantities, profits, consumer surplus, overall public goods provision and welfare for the high cost country in both Stages of the game, depending on whether countries are in autarky or engage in trade during Stage 1.

Tables 1 and 2 provide an overview of prices, quantities, consumer surplus, profits, public good provision and overall welfare in each country, respectively, for both Stages of the game,

depending on which strategy is chosen in Stage 1. It is clear that Stage 1 welfare in both countries is greater under trade, while the high cost country's and collective Stage 2 welfare are greater following autarky in Stage 1.²⁴ Each government must therefore trade off the welfare losses from temporary autarky against the welfare gains derived from the low cost firm's ability to catch up. In the following analysis, I decompose welfare into three parts: consumer surplus; profits; and the public good. I start with overall consumption of the new technology, ie the public good.

In order for $r_i + r_j = R$ to be greater under temporary autarky than under trade, it must be the case that

$$a - \frac{c(1+d)}{2} + 2(a-c) > 4(a-dc)$$
(28)

This holds for any $d > \frac{2a}{7c} + \frac{5}{7}^{25}$ and can be consistent with equation 17 for any d > 1 and c < a.

Result 1. If $d > \frac{2a}{7c} + \frac{5}{7}$, then $R_1 + R_2$ is greater under Stage 1 autarky than under Stage 1 trade.

Turning to consumer surplus, consumer surplus over both periods is greater for country j if

$$\frac{(a-dc)^2}{8} + \frac{(a-c)^2}{2} > (a-dc)^2$$
(30)

This holds for $d > \frac{(\sqrt{7}-2)a+2c}{\sqrt{7}c}$.

Result 2. If $d > \frac{(\sqrt{7}-2)a+2c}{\sqrt{7}c}$, then $\Sigma(CS_{j,1}+CS_{j,2})$ is greater under Stage 1 autarky than under Stage 1 trade.

For country *i* this is the case if

$$\frac{(a-c)^2}{8} + \frac{(a-c)^2}{2} > (a-dc)^2$$
(31)

which holds for $d > \frac{(\sqrt{8}-\sqrt{5})a+\sqrt{5}c}{\sqrt{8}c}$.

Result 3. If $d > \frac{(\sqrt{8}-\sqrt{5})a+\sqrt{5}c}{\sqrt{8}c}$, then $\Sigma(CS_{i,1}+CS_{i,2})$ is greater under Stage 1 autarky than under Stage 1 trade.

Note that Result 1 implies that both Results 2 and 3 also hold. The converse is not the case, implying that there are values of d for which one or both countries would choose temporary autarky over trade, even though the latter would be preferable for public goods provision and collective welfare. Further,

25. This follows from

$$a - \frac{c}{2} - \frac{dc}{2} + 2a - 2c > 4a - 4dc$$

$$\rightarrow d > \frac{2a}{7c} + \frac{5}{7}$$
(29)

^{24.} The low cost country's welfare may be higher or lower in Stage 2 following autarky, depending on whether the increases in consumer surplus and public goods provision are greater or lower than the loss in rents extracted from the high cost country's consumers.

$$\frac{(\sqrt{7}-2)a+2c}{\sqrt{7}c} > \frac{(\sqrt{8}-\sqrt{5})a+\sqrt{5}c}{\sqrt{8}c}$$
(32)

implying that the low cost country enjoys gains in consumer surplus from temporary protection at lower levels of d than the high cost country. Put differently, the high cost country makes a greater temporary sacrifice in terms of consumer surplus by choosing autarky.

The final outcome affected by the choice to engage in trade are profits. Since the high cost firm is inactive under Stage 1 trade, profits in the high cost country will always be greater under autarky. In the low cost country, temporary autarky benefits the producer if

$$\frac{(a-c)^2}{4} > 4c(a-dc)(d-1)$$

$$\rightarrow \frac{(a-c)^2}{4} - 4c(a-dc)(d-1) > 0$$
(33)

Equation 33 holds for any (a) $d > \frac{2a+2c+\sqrt{3}a-\sqrt{3}c}{4c}$ and (b) $d < \frac{2a+2c-\sqrt{3}a+\sqrt{3}c}{4c}$. (a) is not consistent with a d > 1 and a > c.²⁶ (b) is consistent with d > 1 and a > c.²⁷

Result 4. If $d < \frac{2a+2c-\sqrt{3}a+\sqrt{3}c}{4c}$, then the low-cost firm's profits are higher under temporary autarky than under free trade.

However, result 3 and 4 cannot hold at the same time. Proof:

$$\frac{2a+5c}{7c} < d < \frac{2a+2c-\sqrt{3}a+\sqrt{3}c}{4c} \to 4(2a+5c) < 7(2a+2c-\sqrt{3}a+\sqrt{3}c) \to (8-14+7\sqrt{3})a < (14-20+7\sqrt{3})c \to a < c$$
(36)

is not consistent with a > c.

26. This follows from

$$d > 1 > \frac{2a + 2c + \sqrt{3}a - \sqrt{3}c}{4c}$$

$$\rightarrow c > a + \frac{\sqrt{3}}{2}(a - c)$$
(34)

which is not consistent with a > c. 27. Proof:

$$1 < d < \frac{2a + 2c - \sqrt{3}a + \sqrt{3}c}{4c}$$

$$\rightarrow 4c < 2a + 2c - \sqrt{3}a + \sqrt{3}c$$

$$\rightarrow (2 - \sqrt{3})c < (2 - \sqrt{3})a$$
(35)

which implies a > c.





Consumer Surplus, Consumption, Profit and Welfare Under Trade Relative to Autarky





Consumer Surplus, Consumption, Profit and Welfare Under Trade Relative to Autarky





Consumer Surplus, Consumption, Profit and Welfare Under Trade Relative to Autarky

B TRADE AND AUTARKY WITH SUBSIDIES

B.1 Consumer Subsidies

Under autarky, in any Stage of the game each government will subsidize in order to eliminate the deadweight loss from the monopoly and internalize the domestic part of the climate externality. Consider a downstream subsidy in country i. A quantity subsidy s shifts consumer demand such that

37

$$r_i = a - P + s_i \tag{37}$$

The monopolist now maximizes

$$\Pi_{i} = (a - P + s_{i})(P - c)$$

$$\rightarrow P = \frac{a + s_{i} + c}{2}$$
(38)

thus

$$r_i = \frac{a+s_i-c}{2} \tag{39}$$

The government thus faces the objective function

$$W_{i}(s) = ar_{i} - \frac{r_{i}^{2}}{2} - (P - s_{i})r_{i} + Pr_{i} - cr_{i} + \frac{b}{2}(r_{i} + r_{j}) - s_{i}r_{i}$$

$$= (a - c + \frac{b}{2})r_{i} - \frac{r_{i}^{2}}{2} + \frac{b}{2}r_{j}$$

$$= (a - c + \frac{b}{2})(\frac{a + s_{i} - c}{2}) - \frac{(\frac{a + s_{i} - c}{2})^{2}}{2} + \frac{b}{2}r_{j}$$
(40)

From the first order condition we get

$$s_i = a + b - c \tag{41}$$

which yields

$$r_i = a - c + \frac{b}{2} \tag{42}$$

Country *j* solves a symmetric problem, substituting dc for *c*. Quantities and welfare are thus the same as under perfect competition - see appendix C.

Under trade, again assuming that dc is below the low cost country's monopoly price, only the low cost firm is active and supplies the market at P = dc. The low cost country's government sets a subsidy to optimize

$$W_{i} = ar_{i} - \frac{r_{i}^{2}}{2} - (dc - s_{i})r_{i} + (dc - c)(r_{i} + r_{j}) - r_{i}s_{i} + \frac{b}{2}(r_{i} + r_{j})$$
(43)

Given that only $r_i = a - dc + s_i$ and r_j is not affected by the domestic subsidy, the first order condition implies that

$$s_i = c(d-1) + \frac{b}{2} \tag{44}$$

The high cost country's government solves

$$maxW_{j} = ar_{j} - \frac{r_{j}^{2}}{2} - dcr_{j} + \frac{b}{2}(r_{i} + r_{j})$$
(45)

yielding

$$s_j = \frac{b}{2} \tag{46}$$

These subsidies lead to quantities

Result 5. $r_i = a - c + \frac{b}{2}; r_j = a - dc + \frac{b}{2}$

B.2 Producer Subsidies

Autarky Under autarky and upstream subsidies, the firm optimizes

$$\Pi_i = Pr_i - cr_i + s_i r_i$$

= $(P - c + s_i)(a - P)$ (47)

which yields $P = \frac{a+c-s_i}{2}$ and $r = \frac{a-c+s_i}{2}$. The optimal subsidy resulting from the government's maximization problem is $s_i = a + b - c_i$, as it was under downstream subsidies. Government *j* solves a symmetrical problem. Outcomes are identical to perfect competition and autarky (see appendix C).

Trade – Levelled Industry Suppose government *i* changes strategy unilaterally and sets $s_i = b + \varepsilon$. It can now capture the whole market. However, since $s_j = b$, profits effectively remain 0. The government's welfare now becomes

$$W_{i}(b+\varepsilon,b) = CS(b+\varepsilon,b) + \Pi(b+\varepsilon,b) + \frac{b}{2}R(b+\varepsilon,b) - (b+\varepsilon)R(b+\varepsilon,b)$$

$$= CS(b+\varepsilon,b) + 0 + \frac{b}{2}R(b+\varepsilon,b) - (b+\varepsilon)R(b+\varepsilon,b)$$
(48)

which reduces welfare by $\frac{R}{2}(b+\varepsilon)$ – the amount by which it subsidizes country *j*'s consumption. This is clearly not a desirable move.

Conversely, suppose government *i* instead sets $s_i = b - \varepsilon$, such that firm *j* captures the market. Welfare in country *i* becomes

$$W_i(b-\varepsilon,b) = CS(b-\varepsilon,b) + \frac{b}{2}R(b-\varepsilon,b)$$
(49)

while welfare in country j is

$$W_j(b-\varepsilon,b) = CS(b-\varepsilon,b) + 0 + \frac{b}{2}R(b-\varepsilon,b) - bR(b-\varepsilon,b)$$
(50)

s = b is not a sustainable outcome, because each government has an incentive to incrementally reduce subsidies to shift the burden of public goods provision on the other country while maintaining marginal cost pricing. Up to what point? Suppose government *i* expects all production to take place domestically, while price is determined by the subsidy set by government j. $P = c - s_j$, $r_i = r_j = r = a - c + s_j$. The government's problem becomes

$$\max_{\{s_i\}} W_i(s_i, s_j) = \max_{\{s_i\}} CS(s_i, s_j) + \Pi(s_i, s_j) + \frac{b}{2}R(s_i, s_j) - s_i \frac{R(s_i, s_j)}{2}$$
(51)

Setting $u(r_i) = ar_i - \frac{r_i^2}{2}$, we get $s_j = \frac{b+c-a}{3}$. This is the degree to which country *i* is willing to subsidize country *j*, and vice versa.

Result 6. Since it is in country j's interest to maintain marginal cost pricing, the Nash equilibrium in pure strategies is at $s_i = s_j = s = \frac{b+c-a}{3}$ and $P = c - s = \frac{a+2c-b}{3}$ – the same as under perfect competition.

$$\max_{\{s_i\}} W_i(s_i, s_j) = \max_{\{s_i\}} ar_i - \frac{r_i^2}{2} - Pr + PR - cR + s_iR + \frac{b}{2}R - s_iR$$

$$= \max_{\{s_i\}} (a - c + s_j)r - \frac{r^2}{2} + (c - s_j)R - cR + s_iR + \frac{b}{2}R - s_iR$$
(52)

We see all terms containing s_i cancel out. Reframe the problem as

$$\max_{\{s_j\}} W_i(s_i, s_j) = \max_{\{s_j\}} (a - c + s_j)r - \frac{r^2}{2} - s_j R + \frac{b}{2}R$$

$$= \max_{\{s_j\}} \frac{(a - c + s_j)^2}{2} + 2(\frac{b}{2} - s_j)(a - c + s_j)$$
(53)

Trade – Unlevelled Industry Market sharing requires that $dc - s_j = c - s_i$, implying that $s_j = dc + s_i - c$.

However, we have already shown that the largest subsidy which country *i* will support is $s_i = \frac{b+c-a}{3}$, while for country *j* with cost dc this is now $s_j = \frac{b+dc-a}{3}$. $s_j = dc + s_i - c = dc - c + \frac{b+c-a}{3} = \frac{3dc-2c+b-a}{3} > \frac{b+dc-a}{3}$. Country *j* will not be prepared to sustain this subsidy, unless it does not actually expect to pay it. The low cost firm will thus supply the entire market, and its government subsidize both countries' consumers via upstream subsidies. The low cost firm sets $P = dc - s_j$ and $r_i = r_j = r = a - dc + s_j$. Government *i* optimizes the following objective function:

$$W_{i} = (a - dc + s_{j} - \frac{r}{2})r + (dc - s_{j} - c + s_{i})R + \frac{b}{2}R - s_{i}R$$
(54)

Any terms containing s_i cancel out, because price is affected only by s_j , not s_i , and only r_js_j spills over to the other country. The optimal level of s_j from the high cost government's perspective is infinite as long as in practice, its firm is inactive. From the perspective of the low cost government, the first order condition yields the optimal subsidy $s_j = \frac{3dc-2c+b-a}{3}$. This is the highest subsidy which the low cost government is prepared to outbid. If the industry is levelled such that d = 1, this reduces to $\frac{b+c-a}{3}$ – the same subsidy we found in the previous paragraph.

Result 7. In the unlevelled industry, government *j* will therefore set $s_j = \frac{3dc-2c+b-a}{3} - \varepsilon$. Government *i* will set $s_i = \frac{b+c-a}{3}$ and the low cost firm supplies the whole market at $P = dc - s_j = \frac{a+2c-b}{3} + \varepsilon$.

Temporary Production Subsidy Suppose the high cost country subsidizes its industry during Stage 1 in order to catch up with the low cost industry. For simplicity, assume this is the only policy instrument used. In order for the high cost firm to compete, $c = dc - s_j \rightarrow s_j = c(d-1)$.

TABLE 3 Outcome Matrix for Country i

gy Stage 1 Outcome, country <i>i</i> Stage 2 Outcome, country <i>i</i>	$S_{i} = a + b - c; P_{i} = \frac{2c - b}{2}; r_{i} = a - c + \frac{b}{2}$ $S_{i} = \frac{b + c - a}{3}; P_{i} = \frac{a + 2c - b}{3}; r_{i} = \frac{2a - 2c + b}{3}$ $W_{i} = \frac{3}{8}(2a - 2c + b)^{2} + \frac{b}{2}(2a - c(1 + d) + b) - (a + b - c)(a - c + \frac{b}{2})$ $W_{i} = \frac{(2a - 2c + b)^{2}}{18} + \frac{b}{3}(2a - 2c + b) - \frac{b + c - a}{3}\frac{2a - 2c + b}{3}$ $CS_{i} = \frac{(2a - 2c + b)^{2}}{8}; \Pi_{i} = \frac{(2a - 2c + b)^{2}}{4}; B_{i} = \frac{b}{2}(2a - c(1 + d) + b)$ $CS_{i} = \frac{(2a - 2c + b)^{2}}{8}; \Pi_{i} = \frac{(2a - 2c + b)^{2}}{4}; B_{i} = \frac{b}{2}(2a - c(1 + d) + b)$	$s_{i} = \frac{b+c-a}{3}; P_{i} = \frac{a+2c-b}{3}; r_{i} = r_{i} = \frac{2a-2c+b}{3}; r_{i} = r_{i} = \frac$
Stage 1 Strategy	Autarky $W_i = C$	Trade

countries are using producer subsidies, depending on whether countries are in autarky or engage in trade during Stage 1.

Now P = c; $r_i = r_j = a - c$; R = 2(a - c). Both firms supply 50% of the market. Country *j*'s welfare is

$$W_{j} = CS_{j} + \Pi_{j} + B_{j} - s_{j}r_{j}$$

$$= \frac{(a-c)^{2}}{2} + (a-c)(c-dc+c(d-1)) + \frac{b}{2}2(a-c) - c(1-d)(a-c)$$

$$= \frac{(a-c)^{2}}{2} + b(a-c) - c(d-1)(a-c)$$
(55)

country *i*'s welfare is

$$W_{i} = CS_{i} + \Pi_{i} + B_{i}$$

$$= \frac{(a-c)^{2}}{2} + 0 + b(a-c)$$
(56)

In Stage 2, both firms produce at marginal cost *c*, and welfare is $W_i = W_j = \frac{(a-c)^2}{2} + b(a-c)$. Can this be an equilibrium? Assume for the moment that a climate subsidy is not available and compare each country's welfare to its outcomes under trade and autarky with no production subsidy.

In order for country j to prefer trade with a production subsidy over trade without it, it must be the case that

$$(a-c)^{2} + 2b(a-c) - c(d-1)(a-c) > (a-dc)^{2} + 2b(a-dc)$$

$$\to (a-c)(a-c+2b-dc+c) > (a-dc)(a-dc+2b)$$

$$\to a-c > a-dc$$
(57)

This holds for all d > 1. Country *j* will always prefer trade with a production subsidy over trade without it.

What about autarky? Stage 2 outcomes will be identical, so in order for a production subsidy to be preferred welfare in Stage 1 must be greater under trade and a production subsidy. This requires

$$\frac{(a-c)^2}{2} + b(a-c) - c(d-1)(a-c) > \frac{3(a-dc)^2}{8} + \frac{b}{2}\left(\frac{a-dc}{2} + \frac{a-c}{2}\right)$$
(58)

However, country *j* can increase its domestic welfare by reducing the subsidy to $s_j = c(d - 1) - \varepsilon$. This will eliminate the cost of the subsidy, while maintaining marginal cost pricing, and constitutes a pareto improvement. If production subsidies are available in both periods, the equilibrium will not support a 'catching up' subsidy.

C PERFECTLY COMPETITIVE FIRMS

Trade In the absence of subsidies, in any Stage of the game $P_i = P_j = c$, $r_i = r_j = a - c$, R = 2(a - c). The respective countries' welfare is

$$W_{i} = W_{j} = ar_{i} - \frac{r_{i}^{2}}{2} - cr_{i} + \frac{b}{2}R$$

= $\frac{(a-c)^{2}}{2} + b(a-c)$ (59)

Autarky Under autarky without subsidies, $P_i = c$ and $P_j = dc$. $r_i = a - c$, $r_j = a - dc$, R = 2a - c(1 + d). The respective countries' welfare is

$$W_{i} = ar_{i} - \frac{r_{i}^{2}}{2} - P_{i}r_{i} + \frac{b}{2}R$$

$$= \frac{(a-c)^{2}}{2} + ba - \frac{bc}{2}(1+d)$$

$$W_{j} = ar_{j} - \frac{r_{j}^{2}}{2} - P_{j}r_{j} + \frac{b}{2}R$$

$$= \frac{(a-dc)^{2}}{2} + ba - \frac{bc}{2}(1+d)$$
(60)

Since both countries' welfare is higher under trade than autarky, and as there is no benefit to be derived from the high country catching up under marginal cost pricing, it will be welfareimproving for both countries to trade in both Stages of the game.

Autarky With Subsidies Under autarky, each country takes into account only its own share of the positive externality when setting subsidies, such that $s = \frac{b}{2}$. From the respective consumers' demand functions we get

$$r_{i} = a - c + \frac{b}{2}$$

$$r_{j} = a - dc + \frac{b}{2}$$

$$R = 2a + b - c(1 + d) < 2(a - c + b)$$
(61)

Welfare is

$$W_{i} = ar_{i} - \frac{r_{i}^{2}}{2} - cr_{i} + \frac{b}{2}R$$

$$= (a - c) * (a - c + \frac{b}{2}) - \frac{(a - c + \frac{b}{2})^{2}}{2} + \frac{1}{2}(2a + b - c(1 + d))$$

$$W_{j} = ar_{j} - \frac{r_{j}^{2}}{2} - dcr_{j} + \frac{b}{2}R$$

$$= (a - dc) * (a - dc + \frac{b}{2}) - \frac{(a - dc + \frac{b}{2})^{2}}{2} + \frac{1}{2}(2a + b - c(1 + d))$$
(62)

Trade With Consumer Subsidies Under trade and perfect competition, only the low cost firm will be active and provide any quantity at P = c. If governments choose a consumer

subsidy, each government will subsidize at $s = \frac{b}{2}$. Both representative consumers purchase R at the same price $P = c - \frac{b}{2}$, resulting in

$$r_i = r_j = a - c + \frac{b}{2}$$

$$R = 2(a - c) + b$$
(63)

As before, there is no benefit to catching up and both countries wish to trade in both periods.

Trade With Producer Subsidies If governments subsidize their upstream industry instead, they can influence not only their domestic consumers' demand, but also the prices faced by foreign consumers. Assuming that the high cost country does not have considerably higher subsidies than the low cost country (which cannot be an equilibrium, as previous analysis has shown), consumers in both countries face $P = c - s_i$. The low cost firm supplies the entire market such that $q_i = R$. The cost of the subsidy to government *i* is $s_i * R$.

$$P = c - s_i$$

$$r_i = r_j = a - c + s_i$$

$$R = 2(a - c + s_i)$$
(64)

government i will choose s_i to solve

$$\max_{\{s_i\}} W_i = ar_i - \frac{r_i^2}{2} - Pr_i + PR - cR + s_iR + \frac{b}{2}R - s_iR$$
(65)

From the first order condition we obtain the optimal upstream subsidy s_i :

$$s_i = \frac{b - a + c}{3} \tag{66}$$

Inserting the subsidy into consumer demand yields

$$r_{i} = r_{j} = \frac{2a - 2c + b}{3}$$

$$R = \frac{2}{3}(2a - 2c + b)$$
(67)

Note that subsidies, consumption and public goods provision are lower under producer than consumer subsidies. Note also that the high cost country enjoys higher welfare than the low cost country, as the latter bears the full cost of public goods provision. The low cost country may therefore prefer to remain in autarky during both periods.